

### Sensitivity of GPS tropospheric estimates to mesoscale convective systems in West Africa

Samuel Nahmani, Olivier Bock, Françoise Guichard

### ▶ To cite this version:

Samuel Nahmani, Olivier Bock, Françoise Guichard. Sensitivity of GPS tropospheric estimates to mesoscale convective systems in West Africa. Atmospheric Chemistry and Physics, 2019, 19 (14), pp.9541-9561. 10.5194/acp-19-9541-2019. hal-02416194

### HAL Id: hal-02416194 https://hal.science/hal-02416194

Submitted on 2 Nov 2020

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution - NoDerivatives 4.0 International License



# Sensitivity of GPS tropospheric estimates to mesoscale convective systems in West Africa

Samuel Nahmani<sup>1,2</sup>, Olivier Bock<sup>1,2</sup>, and Françoise Guichard<sup>3</sup>

<sup>1</sup>Institut de physique du globe de Paris, Université de Paris, CNRS, IGN, 75005 Paris, France <sup>2</sup>ENSG-Géomatique, IGN, 77455 Marne-la-Vallée, France <sup>3</sup>CNRM, CNRS UMR 3589 and Météo-France, Toulouse, 31057 CEDEX 1, France

Correspondence: Samuel Nahmani (nahmani@ipgp.fr)

Received: 28 November 2018 – Discussion started: 6 March 2019 Revised: 20 June 2019 – Accepted: 28 June 2019 – Published: 29 July 2019

Abstract. This study analyzes the characteristics of GPS tropospheric estimates (zenith wet delays - ZWDs, gradients, and post-fit phase residuals) during the passage of mesoscale convective systems (MCSs) and evaluates their sensitivity to the research-level GPS data processing strategy implemented. Here, we focus on MCS events observed during the monsoon season of West Africa. This region is particularly well suited for the study of these events due to the high frequency of MCS occurrences in the contrasting climatic environments between the Guinean coast and the Sahel. This contrast is well sampled with data generated by six African Monsoon Multidisciplinary Analysis (AMMA) GPS stations. Tropospheric estimates for a 3-year period (2006-2008), processed with both the GAMIT and GIPSY-OASIS software packages, were analyzed and intercompared. First, the case of a MCS that passed over Niamey, Niger, on 11 August 2006 demonstrates a strong impact of the MCS on GPS estimates and post-fit residuals when the GPS signals propagate through the convective cells as detected on reflectivity maps from the MIT C-band Doppler radar. The estimates are also capable of detecting changes in the structure and dynamics of the MCS. However, the sensitivity is different depending on the tropospheric modeling approach adopted in the software. With GIPSY-OASIS, the high temporal sampling (5 min) of ZWDs and gradients is well suited for detecting the small-scale, short-lived, convective cells, while the post-fit residuals remain guite small. With GAMIT, the lower temporal sampling of the estimated parameters (hourly for ZWDs and daily for gradients) is not sufficient to capture the rapid delay variations associated with the passage of the MCS, but the post-fit phase residuals clearly reflect the presence of a strong refractivity anomaly. The results are generalized with a composite analysis of 414 MCS events observed over the 3-year period at the six GPS stations with the GIPSY-OASIS estimates. A systematic peak is found in the ZWDs coincident with the cold pool crossing time associated with the MCSs. The tropospheric gradients reflect the path of the MCS propagation (generally from east to west). This study concludes that ZWDs, gradients, and postfit phase residuals provide relevant and complementary information on MCSs passing over or in the vicinity of a GPS station.

### 1 Introduction

The American Meteorological Society (2015) defines a mesoscale convective system (MCS) as "a cloud system that occurs in connection with an ensemble of thunderstorms and produces a contiguous precipitation area on the order of 100 km or more in horizontal scale in at least one direction. A MCS exhibits deep, moist convective overturning contiguous with or embedded within a mesoscale vertical circulation that is at least partially driven by the convective overturning". Figure 1 displays a typical nighttime example of the convective activity over West Africa during the monsoon season, with two active clusters of MCS on the eastern and western borders of Nigeria, and more scattered dissipating MCSs over Guinea. Small isolated developing MCSs can also be seen in northern Ghana and over the Senegal-Mauritania border. The MCS's internal structure typically comprises convective and stratiform regions as described by Zipser (1977),



**Figure 1.** Meteosat-8 infrared image of West Africa on 10 August 2006 at 23:00 UTC. Brightness temperatures range between 225 K (yellow) and 195 K (red). The position of the six AMMA GPS stations are shown as white dots: TOMB (Timbuktu, Mali), GAO1 (Gao, Mali), OUAG (Ouagadougou, Burkina Faso), NIAM (Niamey, Niger), TAMA (Tamale, Ghana), and DJOU (Djougou, Benin). Meteosat-8 infrared background: © 2019 EUMETSAT.

Houze et al. (1990), and Houze (1989, 1997, 2004). However, numerous questions related to MCSs remain poorly answered, such as the mechanisms behind their life cycle (their initialization, propagation, and decay; e.g., Laing et al., 2008), and how these mechanisms involve interactions between physical processes (e.g., surface and boundary layer processes – Taylor et al., 2017; Rochetin et al., 2017) and atmospheric circulations (e.g., African easterly waves – Fink et al., 2006; Rickenbach et al., 2009; Beucher et al., 2014).

Deep convection, MCSs, and rainfall are strongly linked to atmospheric water vapor (e.g., Bretherton et al., 2004; Schiro et al., 2016; Sherwood et al., 2010). This is why GPS integrated water vapor (IWV) data have been used for studies in various tropical climates such as the Indian monsoon (Jade et al., 2005; Jade and Vijayan, 2008), the South and North American monsoon (Gutzler et al., 2004; Means, 2013; Adams et al., 2011, 2014; Serra et al., 2016), and the West African monsoon (WAM) (Couvreux et al., 2012; Barthe et al., 2010). Here, we focus on MCSs occurring during the WAM, which has been the subject of intense investigations in recent years in the framework of the African Monsoon Multidisciplinary Analysis (AMMA) program (Redelsperger et al., 2006; Lafore et al., 2011). Indeed, MCSs are a key component of the WAM, as they produce 90% of the annual rainfall in West Africa (D'Amato and Lebel, 1998). Within the AMMA's framework, six GPS stations (Fig. 1) were operated between 2005 and 2010 to sample IWV along the climatic gradient from the arid Saharo-Sahelian climate in Mali to the moist Sudano-Guinean climate in Benin (Bock et al., 2008). The AMMA GPS data have proved to be valuable for understanding the water cycle of this region including WAM atmospheric processes (Janicot et al., 2008; Lafore et al., 2011), land water storage changes (Hinderer et al., 2009), water budgets (Meynadier et al., 2010a, b), or hydrological loading deformation induced by the WAM (Nahmani et al., 2012).

Links between atmospheric water vapor and convective precipitation occur on various temporal scales down to less than a few hours. Notably, GPS data have revealed that a sharp IWV increase is often observed with the passage of MCSs (Barthe et al., 2010; Schiro et al., 2016; Taylor et al., 2017). In this study, we focus on this "small" scale (a few hours or less). More precisely, we address the question of the accuracy of AMMA GPS observations and the informational content of tropospheric delay estimates during the passage of MCSs over West Africa, comparing the two main researchlevel GPS data processing approaches: network processing, in which differentiated or undifferentiated observations from several stations are processed simultaneously, and precise point positioning (PPP), in which each station is processed independently but requires precise satellite orbits and clocks as calculated by the International GNSS Service (IGS) or one of its analysis centers (http://www.igs.org, last access: 25 July 2019). In both approaches, many processing options can be used, depending on the functionality of the software, such as least squares adjustment or Kalman filter, the choice of the elevation cutoff angle, the modeling of the troposphere, the rejection procedure of the erroneous observations (screening), among others. In this paper we compare the GPS Analysis at Massachusetts Institute of Technology (GAMIT) software (Herring et al., 2015) and the GNSS Inferred Positioning System and Orbit Analysis Simulation (GIPSY-OASIS) software from the NASA Jet Propulsion Laboratory (JPL)/Caltech (https://gipsy-oasis.jpl.nasa.gov/, last access: 25 July 2019). The first performs a least squares adjustment on a regional network with one zenith tropospheric delay (ZTD) estimated every hour, whereas the second uses a Kalman filter to process the data in PPP mode with one ZTD estimated every 5 min. Both software packages implement tropospheric delay models that also include gradient parameters. Gradient parameters allow for the representation of the effect of first-order azimuthal asymmetry in the atmospheric refractive index (MacMillan, 1995; MacMillan and Ma, 1997; Chen and Herring, 1997). The spatial scale over which the GPS measurements are sensitive to atmospheric refractivity gradients is about 50 km.

Section 2 discusses the processing options available in both software packages and the expected impact on the tropospheric delay estimates. Results are presented and discussed for the AMMA GPS stations over a continuous period of 3 years (2006 to 2008 – the most complete period for the six stations). The climatic context of West Africa lends itself particularly well to this methodological study with marked dry and wet seasons. Section 3 presents the case study of an organized cluster of MCSs passing over Niamey, Niger, on 11 August 2006 (Fig. 1). The case study illustrates the main characteristics of the MCS events captured by GPS tropospheric estimates according to the tropospheric modeling used during the data processing. Section 4 provides at statistical characterization of MCSs using GPS tropospheric delay estimates from 414 events observed by the six AMMA stations between 2006 and 2008. Section 5 concludes the study.

#### 2 GPS data processing and methods

The six AMMA GPS stations used here meet the quality standards of the IGS (Dow et al., 2009) and are also equipped with meteorological sensors (Vaisala, PTU200). Their Receiver-Independent Exchange (RINEX) files are freely available on IGS servers (see the data availability section of this paper).

### 2.1 GPS data analysis with GAMIT

The RINEX data from the six AMMA GPS stations are processed from 1 January 2006 to 31 December 2008 within a regional network of around 25 IGS stations using the GAMIT scientific software (Herring et al., 2015) release 10.6. The settings of the GPS data processing are designed to meet the recommendations of the International Earth Rotation Service (IERS) conventions (Petit and Luzum, 2010) and correspond to an update of those used by Bock et al. (2008). The IGS final orbits are held fixed. A priori station positions expressed in ITRF2014 (Altamimi et al., 2016) are constrained with 10 and 30 cm a priori standard deviations for the horizontal and vertical components, respectively. GPS phase observations are corrected by applying absolute antenna phase calibration models (Schmid et al., 2007) and are weighted using the elevation-dependent variance function  $\sigma^2 = a^2 + b^2 / \sin^2(e)$ , where e is the elevation angle and a and b are determined by a least squares of the post-fit residuals in an iterative procedure. The integer ambiguities are determined using wadelane and narrow-lane combinations (Dong and Bock, 1989). The elevation cutoff angle is fixed to  $7^{\circ}$ . Ocean tide loading effects are corrected using the FES2004 model (Lyard et al., 2006). Atmospheric loading and non-tidal ocean loading can be neglected at AMMA GPS stations (Nahmani et al., 2012). This approximation introduces an uncertainty at a submillimetric level on zenith wet delay (ZWD) estimates at AMMA GPS stations (Nahmani, 2012). Concerning the propagation delays, the ionosphere-free phase combinations are corrected for second- and third-order ionospheric refraction terms (Hernandez-Pajares et al., 2007; Petrie et al., 2010). For tropospheric modeling, we use the VMF1 mapping functions and a priori hydrostatic delays derived from the European Centre for Medium-Range Weather Forecasts (ECMWF) model (Boehm et al., 2006b). The ZWDs and total tropospheric horizontal gradients are estimated with sampling intervals of 1 and 24 h, respectively. During the least squares estimation, the ZWD is represented by a piecewise linear function with hourly nodes modeled as a random walk, i.e., ZWD( $t + \Delta t$ ) = ZWD(t) +  $\varepsilon(t)$ . The variance of the white noise  $\varepsilon(t)$  is parameterized as  $\sigma_{\varepsilon}^2 = q_{rw}^2 \cdot \Delta t$ , where  $\Delta t = 1$  h, and the parameter  $q_{rw}$  is fixed to the recommended value of 20 mm h<sup>-1/2</sup> (Herring et al., 2015). To avoid edge effects on the ZWD estimates, we use a sliding window technique. We process GPS data in two 24 h sessions, the first starting at 00:00 UTC and the second at 12:00 UTC, from which the central 12 h output estimates are extracted as the final solutions. At the end of each session processing, the estimated tropospheric delays, tropospheric gradients, and the GPS one-way phase residuals (Herring et al., 2015) are saved for each AMMA station (Table 1).

The processing of GPS data with the GAMIT software has some limitations: the ZWD temporal sampling interval of 1 h and the gradient temporal sampling interval of 24 h are too coarse to document the rapid meteorological extreme events like the passage of MCSs. Indeed, in such situations, rapid variations in the ZWD and strong anisotropy in the refractivity field are observed (Bock et al., 2008). As a consequence of the severe mis-modeling of the propagation delays, the estimated ZWD and gradient parameters are likely biased. Inspection of the post-fit phase residuals will help identify the severity of the mis-modeling. This mis-modeling can be partly overcome via the use of a PPP Kalman-filter-based software to increase the sampling interval of ZWD and gradient estimates. The PPP approach also has the advantage of eliminating the propagation of errors induced by the mismodeling between stations.

### 2.2 GPS data analysis with GIPSY-OASIS in PPP mode

The GIPSY-OASIS II v 6.3 software is based on a Kalmanfilter estimator (Zumberge et al., 1997). We used it in PPP mode to reprocess the AMMA GPS data with a parameterization that is as similar as possible to the settings used with the GAMIT software described in Sect. 2.1. Table 1 summarizes the different strategies used to process the GPS data. Compared with GAMIT, a major difference with GIPSY-OASIS II is the sampling interval of the estimated parameters (ZWDs, tropospheric gradients, and receiver clocks) of 5 min, which is achievable thanks to the Kalman-filter technique in which the parameters are updated at each time step. A minor drawback is that the GPS phase observations are thus reduced to a 5 min interval, i.e., this estimation procedure uses 10 times fewer observations than the GAMIT processing. A few other minor differences are also found in the observation weighting function, which is modeled as  $\sigma^2 = a^2 / \sin^2(e)$  with a fixed to 10 mm, and in the phase ambiguities resolution methods (GIPSY-OASIS II v6.3 uses the "ambigon" algorithm of Bertiger et al., 2010). To avoid edge effects, we followed the standard procedure used at JPL. The GPS data are analyzed in a 30h window centered on each

	GAMIT v10.6	GIPSY-OASIS II v6.3
Strategy	Network of 25 stations centered on West Africa (6 AMMA stations + 19 regional stations); free network strategy using double-difference baselines; 7° elevation cutoff angle; data processed in two 24 h sessions (from 00:00 to 24:00 UTC and from 12:00 UTC on day D to 12:00 UTC on day $D + 1$ ), from which the cen- tral parts are extracted (06:00–18:00 and 18:00– 06:00 UTC).	Stations processed individually; precise point positioning (PPP) mode using un- differenced phase observations; $7^{\circ}$ elevation cutoff angle; data processed in a single 30 h session (from 21:00 UTC on day $D - 1$ to 03:00 UTC on day D + 1), from which the central part is extracted (00:00-24:00 UTC).
Orbits and clocks	IGS final orbits fixed.	JPL final orbits and clocks fixed (sampled at 5 min dedicated to PPP mode).
Observation rate	30 s sampling interval.	5 min sampling interval.
Observations weighting	$\sigma^2 = a^2 + b^2 / \sin^2(e)$ , where <i>e</i> is the elevation angle and <i>a</i> and <i>b</i> are determined from least squares adjustment residuals.	$\sigma^2 = a^2 / \sin^2(e)$ with <i>a</i> fixed to 10 mm.
Tropospheric modeling	VMF1 mapping function; ZHD (zenithal hydrostatic delays) a priori from ECMWF, one ZWD per hour constrained by a 20 mm h <sup>-1/2</sup> random walk; one tropospheric gradient per 24 h session with- out stochastic constraint.	VMF1 mapping function; ZHD a priori from ECMWF, one ZWD per 5 min constrained by a $20 \text{ mm h}^{-1/2}$ random walk; one tropospheric gradient per 5 min constrained by a 2 mm h <sup>-1/2</sup> random walk.
Tropospheric estimates	One ZWD per hour; one tropospheric gradient per 24 h.	One ZWD per 5 min; one tropospheric gradient per 5 min.

 Table 1. GPS data processing details with the GAMIT and GIPSY-OASIS software packages.

day, from which the 00:00-24:00 UTC parameters are extracted. This is also consistent with the JPL satellite orbit and clock products that are used instead of the IGS products. Regarding tropospheric modeling, the ZWD and tropospheric gradient parameters are modeled as random walk processes with a 5 min time resolution. However, the question then arises as to the choice of the parameterization of the random walk. The recommendations in the GIPSY-OASIS II documentation are mainly based on the results of Bar-Sever et al. (1998) and updated by Selle and Desai (2016). They suggest that the parameter of the ZWD random walk should be set to at least 8.4 mm h<sup>-1/2</sup> when using  $\sigma^2 = a^2/\sin^2(e)$ as the weighting function for the GPS observations (as opposed to  $20 \text{ mm h}^{-1/2}$  in the GAMIT software). However, Jarlemark et al. (1998) noticed that using a random walk parameter tighter than the real atmospheric variability introduces errors in the ZWD estimates. As this study is focused on MCSs which correspond to severe weather events, we can expect a very high variability in the tropospheric parameters; hence, it seems more reasonable to set the random walk parameter to the looser recommended value in both software packages rather than to the tighter. As for the temporal evolution model of tropospheric gradients, it is common to use a random walk for which the parameter is 10 times smaller than that of the ZWD (Bar-Sever et al., 1998). Thus, we keep the GAMIT-recommended random walk parameter of ZWD at 20 mm  $h^{-1/2}$  and set the random walk parameter of the tropospheric gradients to 2 mm  $h^{-1/2}$ .

## 2.3 Comparison and quality of the GPS solutions from the GAMIT and GIPSY-OASIS software packages

### 2.3.1 ZWD and gradients

The time series of ZWDs retrieved by the GAMIT software at Timbuktu, Niamey, and Djougou, between 2006 and 2008, display a well-defined annual cycle (Fig. 2). The dry seasons are characterized by ZWDs of less than 100 mm (ca. between December and March), whereas the wet seasons, also referred to as the monsoon seasons, between June and September (hereafter referred to as JJAS), are characterized by frequent rain events and ZWD values higher than 200 mm (Bock et al., 2008). The duration of the monsoon season shrinks as one goes from the wetter climate in the south to the more arid climate in the north.



Figure 2. Time series of GPS-derived ZWD values retrieved with the GAMIT software and rainfall retrieved from the TRMM Multisatellite Precipitation Analysis (Huffman et al., 2007) at (a) Timbuktu, (b) Niamey, and (c) Djougou. The thin black solid and dashed lines show 30 d moving averages and standard deviations, respectively. The dashed black vertical lines delimit the period between June and September corresponding to the main rainy season.

Table 2 presents the statistics of the differences in the ZWD estimates ( $\Delta$ ZWD) obtained with both software packages during the monsoon season. The average differences of the ZWD values range from -0.45 mm at Gao to 0.96 mm at Djougou. They are small but statistically significant at all stations (95% confidence level) and might be explained by minor differences persisting between the two GPS data processing methods (differences in orbits, tropospheric modeling, weighting of observations, and the potential rejection of some data). Standard deviations of  $\Delta$ ZWD range from 3.69 mm at Tamale to 4.30 mm at Gao. To understand the origin of these differences, we carried out a further statistical analysis of the tropospheric estimates presented in Fig. 3 for ZWDs and in Fig. 4 for gradients at Niamey. Figure 3a shows that the time series of  $\Delta$ ZWD exhibits a greater dispersion during the wet season than during the dry season as expected from larger (smaller) biases and mis-modeling of

**Table 2.** Statistics of ZWD differences from GAMIT and GIPSY-OASIS processing solutions (GAMIT minus GIPSY-OASIS) for June to September, 2006–2008. The GIPSY-OASIS ZWD estimates were averaged over 1 h intervals beforehand. The last column gives the *p* value of a *t* test for the difference of the means. Values smaller than  $1 \times 10^{-7}$  are replaced with zero.

Mean ΔZWD (mm)	SD ΔZWD (mm)	Number of ZWD pairs	p value
0.69	4.03	6874	0
-0.45	4.30	8609	0
0.52	4.17	8588	0
-0.24	4.37	8170	$9.77 \times 10^{-7}$
-0.11	3.69	5731	$2.55 \times 10^{-2}$
0.96	4.18	8720	0
	Mean ∆ZWD (mm) 0.69 -0.45 0.52 -0.24 -0.11 0.96	Mean ΔZWD (mm)         SD ΔZWD (mm)           0.69         4.03           -0.45         4.30           0.52         4.17           -0.24         4.37           -0.11         3.69           0.96         4.18	Mean ΔZWD (mm)         SD ΔZWD (mm)         Number of ZWD pairs           0.69         4.03         6874           -0.45         4.30         8609           0.52         4.17         8588           -0.24         4.37         8170           -0.11         3.69         5731           0.96         4.18         8720

tropospheric refractivity with GAMIT during the wet (dry) season. To better characterize the timescales of atmospheric variability we computed sub-daily and sub-hourly variability estimates. To quantify the sub-daily (i.e., diurnal) variability, we subtracted a 1 d moving average from the hourly time series and then computed daily standard deviations. The sub-hourly variability was quantified in a similar way: by subtracting the hourly time series from the 5 min time series. Figure 3b and c show that the sub-daily and sub-hourly variability in ZWD estimated with GIPSY-OASIS dramatically increases during the wet season. The GAMIT solution presents very similar behavior for the sub-daily variability (not shown), but careful inspection of the 30 d smoothed time series in Fig. 3b shows that the GAMIT solution exhibits slightly larger sub-daily variability than GIPSY-OASIS solution during the wet season. This can be explained by random ZWD biases induced by mis-modeling. Indeed, the tropospheric modeling used in the GAMIT software does not take the sub-daily variations of the gradients into account nor does it account for the sub-hourly variations of ZWD. Therefore, the effect is more pronounced during the wet season because of higher tropospheric variability in these parameters as shown in Figs. 3c and 4 (and further analyzed in a case study in Sect. 3).

Tropospheric gradients show a strong seasonality in phase with the WAM (Fig. 4a, b). The daily averages of gradients are in good agreement between the two software packages (not shown) with a mean difference smaller than a tenth of a millimeter and a standard deviation smaller than 0.22 mm due to biases induced by the un-modeled sub-daily variability. The sub-daily variations of the gradients estimated with GIPSY-OASIS can be quite large (up to 2 mm), with an average standard deviation of around 0.6 mm in both components, during the wet season (Fig. 4c, d). In contrast, the sub-hourly variations of gradients are quite small with an average standard deviation of 0.12–0.15 mm (Fig. 4e, f).

At an elevation of 7°, an uncertainty of the order of 1.5 mm in the ZWD during the JJAS periods (see Fig. 3c for the 1D-



**Figure 3.** Analysis of the variability of GPS-derived ZWD at Niamey. (a) Difference between GAMIT and GIPSY-OASIS hourly ZWD estimates. (b) Sub-daily ZWD variability from GIPSY-OASIS (the daily standard deviation of the hourly minus the daily mean ZWD series is shown in green). (c) Sub-hourly ZWD variability from GIPSY-OASIS (the daily standard deviation of the 5 min minus the hourly ZWD series is shown in green). The thin black solid lines in panels (a), (b), and (c) show the 30 d moving medians. The red line in panel (b) shows a 30 d moving median of the sub-daily variability computed from the ZWD estimates retrieved by GAMIT. The thin black dashed lines in panels (a) and (c) show the median absolute deviations around the medians. Dashed black vertical lines indicate the JJAS period of each year.

STD of "5 min ZWD –  $MA_{1H}$ ") leads to an uncertainty of 12 mm in the slant tropospheric delay (S-Trop-D) using the global mapping function (GMF; Boehm et al., 2006a). Failure to take into account the hourly variations of the gradient of the order of 0.6 mm leads to an uncertainty in the S-Trop-D at an elevation of 7° of around 33 mm using the specific mapping function of Chen and Herring (1997). Neglecting the sub-hourly variations of the gradient (of the order of 0.2 mm) leads to an uncertainty of about 11 mm in



Tropospheric gradients [mm] at Niamey (Niger)

**Figure 4. (a)** North and **(b)** east components of the 5 min tropospheric gradients retrieved with the GIPSY-OASIS software at Niamey between 2006 and 2008, with the 30 d moving averages shown as solid black lines, and the 30 d moving standard deviations around the means shown as dashed black lines. **(c, d)** Sub-daily variability of north and east gradients (similar to Fig. 3b). **(e, f)** Sub-hourly variability of north and east gradients (similar to Fig. 3c). In panels **(c)**–**(f)**, the solid black lines show 30 d moving medians, and the dashed black lines show moving median absolute deviations around the medians. The dashed black vertical lines indicate the JJAS period of each year.

the S-Trop-D. Thus, the global uncertainty in the S-Trop-D induced by these defects of tropospheric modeling in the setting of the GPS data processing with the GAMIT software is evaluated at 37 mm (i.e., 4.7 mm on ZTD if projected back to zenith using the GMF) during the wet season. During DJFM periods, the global uncertainty in the S-Trop-D is evaluated at 16 mm (i.e., 2 mm on ZTD) with 0.9, 0.25, and 0.09 mm as uncertainties in the ZWD, hourly, and subhourly variations of the gradient, respectively. Tropospheric mis-modeling maps onto the GPS phase residuals and its ef-

fects are modulated by the season, which leads to a seasonal variation of the weighted root mean square (WRMS) of the post-fit phase residuals.

### 2.3.2 WRMS of GPS phase residuals

Post-fit phase residuals of the ionosphere-free observations are traditionally inspected to check the quality of GPS observations and models (Kouba, 2003). Usually, the quality of the data processing of each session (24 or 30 h) is summarized in the WRMS of all of the residuals of the session that is computed by the software. With GAMIT, the daily WRMS of phase residuals fluctuates around a median of 12.1 mm with a median absolute deviation (MAD) of 2.4 mm, over the entire period and all stations. Over the same period, with the GIPSY-OASIS software, the median is 10.4 mm with a MAD of 1.1 mm. The statistics are very similar, indicating that the solutions from both software are equivalent on average, although the results suggests that the data processing implemented in GIPSY-OASIS is slightly more accurate.

We know from the previous sections that the mis-modeling of tropospheric variability with GAMIT is exacerbated during the wet season. Figure 5 shows the time series of daily WRMS of post-fit phase residuals at three stations for both software packages. The contrast in WRMS between the wet and dry season is very marked in GAMIT at all three sites (e.g., 12 mm vs. 8 mm on average at Timbuktu). The use of only one tropospheric gradient per 24 h session during the wet season is less suitable for tropical stations; thus, the unmodeled tropospheric anisotropy is mainly reflected by the behavior of the GPS phase residuals. With GIPSY-OASIS, there is also a small seasonal variation, indicating that during the wet seasons some modeling defects may persist (e.g., higher-order tropospheric anisotropy not modeled by tropospheric gradient parameters and signal scattering from the ground, the so called multipath, which depends on soil moisture; e.g., Larson et al., 2008). During the dry season, the WRMS values from GAMIT and GIPSY-OASIS are equivalent at the Sahelian stations (Timbuktu and Niamey in Fig. 5), which indicates that the tropospheric model parameterization used in GAMIT (one ZWD every hour and one gradient every 24 h) is adequate.

Overall, the analysis of the annual cycle presented in Sect. 2 emphasizes a strong increase in the sub-daily variability of GPS tropospheric estimates (ZWD and gradients) and post-fit phase residuals during the wet season, when the atmosphere is moister and MCSs are observed. In the following section the analysis of a case study allows us to shows how such MCSs generate variability at this fine timescale.

S. Nahmani et al.: Sensitivity of GPS tropospheric estimates to mesoscale convective systems



Figure 5. Daily weighted root mean square of the GPS postfit phase residuals at (**a**, **d**) Timbuktu, (**b**, **e**) Niamey, and (**c**, **f**) Djougou. (**a**–**c**) Results of GAMIT software and (**d**–**f**) results of GIPSY-OASIS software. The solid black lines show 30 d moving medians, and the dashed black lines show moving median absolute deviations around the medians.

### 3 Case study of MCS over Niamey, Niger, on 11 August 2006

The MCS that passed over Niamey (Niger) on 11 August 2006 is a typical case of an intense weather event in the Sahel associated with the WAM. It has been the subject of a comprehensive study by Chong (2010) – see also Risi et al. (2010) – and is well-documented by radar and meteorological data. Here, the relevance and limitations of the GPS tropospheric estimates (ZWD and gradients) and post-fit phase residuals during the passage of this MCS are investigated.

The MCS formed over the border between Chad and Nigeria on the evening of 9 August 2006. It propagated westward,



**Figure 6.** Reflectivity maps from the MIT C-band Doppler radar in Niamey at (a) 02:10 UTC, (b) 02:40 UTC, (c) 03:30 UTC, (d) 04:40 UTC, and (e) 05:40 UTC on 11 August 2006, showing a squall-line MCS approaching from the east and passing straight over the site at 03:30 UTC. Post-fit phase residuals from GPS data processing are superposed on the maps using the GIPSY-OASIS software at the exact times of the radar plots. The radar reflectivity color bar is in the lower right-hand panel. The GPS residual color bar is attached to each panel. The GPS phase residuals are projected on the radar maps in the directions of the satellite-receiver paths, assuming a 10 km troposphere thickness and a spherical Earth with a 6400 km radius. The concentric circles indicate the distance from the GPS station's location at 25 km intervals.

intensified over Nigeria on 10 August, and reached Niamey at 03:15 UTC on 11 August. It was observed from satellite imagery (Fig. 1) approaching Niamey from the east. Figure 6 shows reflectivity maps from the MIT C-band Doppler radar at 02:10 UTC (panel a), 02:40 UTC (panel b), 03:30 UTC (panel c), 04:40 UTC (panel d), and 05:40 UTC (panel e) that illustrate the dynamics of the MCS organized into a commonly observed north–south oriented squall-line. The two parts of the MCS are easily identifiable: the convective part with a reflectivity above than 37 dBZ (in yellow, orange, or red), and the stratiform part, which is more extended, with a reflectivity lower than 37 dBZ (mainly in green). The GPS phase residuals superposed in the figure are discussed later in Sect. 3.3.

#### 3.1 Meteorological surface data

Figure 7 shows surface meteorological data retrieved from the Atmospheric Radiation Measurement (ARM) Mobile Facility (ARM-MF) in Niamey (Miller and Slingo, 2007) at



**Figure 7.** Meteorological data during the MCS event on 11 August 2016 at Niamey (Niger) retrieved from the ARM Mobile Facility at a 1 min sampling interval: (a) wind speed (m s<sup>-1</sup>) and direction (°), (b) temperature (°C), (c) relative humidity (%), and (d) pressure (hPa) are shown using black or green lines, and rainfall (mm h<sup>-1</sup>) is shown using blue bar plots. The times for the reflectivity maps given in Fig. 6 are shown using vertical dashed green lines and correspond to (a) 02:10 UTC, (b) 02:40 UTC, (c) 03:30 UTC, (d) 04:40 UTC, and (e) 05:40 UTC.

a 1 min sampling interval (black lines) between 00:00 and 08:00 UTC. These sensors are collocated with the AMMA GPS station at the Niamey airport site. The ARM-MF data are preferentially used for this case study because of their accuracy and their higher temporal frequency.

Time series of wind speed and direction (Fig. 7a) allow for the arrival of the gust-front preceding deep precipitating convective cells to be precisely detected (e.g., Largeron et al., 2015; Lothon et al., 2011). Provod et al. (2016) associated the cold pool crossing time (CPCT) with the time of a sudden wind direction change that occurs within 5 min or less and reaches at least 30° in magnitude. Between 00:00 and 03:03 UTC, the wind speed varies between 1.5 and  $3 \text{ m s}^{-1}$ with a few peaks (not exceeding  $4 \text{ m s}^{-1}$ ) and a direction ranging from 200 to 250°. Between 03:03 and 03:18 UTC, it is around  $2 \text{ m s}^{-1}$ ; it then drops to less than  $1 \text{ m s}^{-1}$  at 03:06 UTC, and suddenly jumps to  $5.8 \text{ m s}^{-1}$  at 03:18 UTC, while the wind direction changes from 200 to 110°. The gustfront is detected at 03:16 UTC (Chong, 2010), and the first rainfall is recorded 18 min later at 03:33 UTC with the arrival of a mature deep convective cell. During this period, the wind speed is significantly higher, ranging from 4 to  $6 \text{ m s}^{-1}$ , it then slightly weakens a few minutes before the first rainfall. The surface temperature rapidly drops from 26.5 to 24.5 °C (Fig. 7b). Similarly, the relative surface humidity suddenly and very briefly drops from 74% to 70% within 2 min at the CPCT (Fig. 7c). This time sequence is very typical of the high-frequency fluctuations of surface meteorology observed during the passage of a MCS (e.g., Zipser, 1977). The relative surface humidity then starts rising again before the first rainfall, exceeding 76 % before rainfall occurs.

From 03:33 to 06:41 UTC, the MCS event is characterized by a two-phase rainfall pattern produced successively by its convective and stratiform parts. Between approximately 03:33 and 04:20 UTC, the convective part of the MCS produces heavy convective showers greater than  $10 \text{ mm h}^{-1}$  over a cumulative period of 29 min and exceeding  $20 \text{ mm h}^{-1}$  over a cumulative period of 12 min. At the beginning of the first shower, a second significant drop in surface temperature, from 24.5 to 22 °C between 03:33 and 03:46 UTC, is observed (Fig. 7b). Likewise, the relative humidity sharply increases to about 92 %. From around 04:20 UTC, the trailing part of the MCS produces stratiform rainfall with a much lower intensity and a much longer duration than convective rainfall: lower than  $1 \text{ mm h}^{-1}$  over a cumulative period of 94 min. The wind speed is then weaker (between 2 and  $4 \text{ m s}^{-1}$ ), before it drops to almost  $0 \text{ m s}^{-1}$ , which, with the cessation of rainfall, marks the end of the MCS event.

Surface pressure also fluctuates during the three major steps of the MCS's lifetime: it increases during the minutes preceding the formation of the convective cell above Niamey; it presents a local maximum during the convective phase; and then it slightly decreases during the stratiform phase. The surface pressure sequence is again consistent with previous studies (e.g., Redelsperger et al., 2002).

There is certainly added value from the high sampling data from the ARM-MF with respect to capturing the details of the internal dynamics of the MCS. However, such data are only available at Niamey (Niger), and only for 2006. The data retrieved from the PTU200 sensor are in good agreement with data from ARM-MF, but the 15 min sampling interval is not sufficient to detect the surface temperature drops (in two consecutive stages) or the sudden and brief drop in relative humidity (not shown). Moreover, because the GPS stations were aimed at providing IWV data, the wind was not recorded. Thus, the best way to identify the CPCT with only PTU200 data at the other AMMA GPS stations is to detect significant drops in surface temperature over a period from 30 min to 1 h before strong rainfall (see Sect. 4).

### 3.2 GPS estimates

The GPS tropospheric estimates retrieved from the GIPSY-OASIS (black lines) software allows for the MCS to be documented with a 5 min sampling rate (Fig. 8). It includes the estimates from the GAMIT software (red dashed lines). In Fig. 8a, the ZWD is retrieved from the ZTD, and the hydrostatic part is computed using 1 min ARM-MF surface pressure data and by applying the formula from Saastamoinen (1972). Figure 8a also shows the IWV converted from the ZWD using 2 m temperature data and the formula from Bevis et al. (1992). East and north components of the tropospheric gradients are given in Fig. 8c and d, respectively.

Comparing the ZWD estimates retrieved from GAMIT and GIPSY-OASIS (Fig. 8a), good agreement can be seen, such as the ZWD peak coinciding with the rainfall event. However, the 5 min sampled ZWD time series from GIPSY-OASIS reveal finer timescale information. ZWD increases from 315 mm at 01:40 UTC to 372 mm at 03:25 UTC, just a few minutes before the first rainfall (Fig. 8a). It remains above 360 mm from 03:00 to 04:10 UTC, while heavy rainfall concurrently dries out the atmosphere from 03:33 UTC onwards. Over this period, the equivalent IWV is quite stable, between 55 and 57 kg m<sup>-2</sup>, while around  $10 \text{ kg m}^{-2}$  of water vapor is extracted from the total column by precipitation (Fig. 8b). The dynamics of the active convective cell involves a low-level convergence of moist air from its immediate environment, which initiates a lift and leads to the significant rainfall amount of 10.5 mm over a 37 min period. As a first approximation, the stability of IWV is well explained by a balance between moisture convergence and precipitation (this implies very minor contributions from surface evaporation and water phase changes to the water budget at this scale). The equilibrium can be broken, e.g., by mixing with drier air in the immediate environment of the cell and/or by processes related to the formation of the incipient convective cells. Such a break occurs around 04:05 UTC - as shown by a decrease in the ZWD followed by a decrease in the precipitation intensity from 04:10 UTC onwards - and marks the transition from the convective part of the MCS to its stratiform part. Between 04:10 and 06:00 UTC, the ZWD decreases from 360 to 297 mm, which corresponds to a decrease of around  $9.7 \text{ kg m}^{-2}$  in the IWV. Over the same period,  $2.6 \text{ kg m}^{-2}$  of atmospheric water is precipitated, which implies that  $7.1 \text{ kg m}^{-2}$  of water has been transported elsewhere, possibly to the newly formed convective cells. Thus, the troposphere is cooler and drier after the passage of the MCS.

Tropospheric gradients retrieved from the GIPSY-OASIS software show the east-west motion dynamic of the MCS between 01:00 and 07:00 UTC. From 01:30 UTC onwards,



**Figure 8.** Tropospheric estimates from the GIPSY-OASIS (black solid line) and GAMIT (red dashed line) software packages during the MCS event on 11 August 2016 at Niamey, Niger: (**a**) zenithal wet delay (mm) and its equivalent IWV (kg m<sup>-2</sup>; red dots), (**c**) east component of the tropospheric gradients, and (**d**) north components of the tropospheric gradients (mm). (**b**) The cumulative rainfall (kg m<sup>-2</sup>; black line) is computed from the 1 min retrieved rainfall (mm h<sup>-1</sup>; blue bar plots) from the ARM Mobile Facility. Rainfall (mm h<sup>-1</sup>; blue bar plots) in panel (**a**) is repeated from (**b**)–(**d**) but is not displayed with a dedicated graduation axis. The times of the reflectivity maps given in Fig. 6 are shown using vertical dashed green lines and correspond to (a) 02:10 UTC, (b) 02:40 UTC, (c) 03:30 UTC, (d) 04:40 UTC, and (e) 05:40 UTC.

the east component increases from 0.26 mm and reaches a maximum of 2.25 mm at 03:20 UTC, whereas the north component remains almost null over this same period. The tropospheric gradient vector actually points to the direction of the incoming convective cells and its modulus shows a first local maximum at the CPCT (Fig. 8c, d). This is the time when the anisotropy of the troposphere around the GPS station is the most pronounced (Fig. 6c). Following this, the east component decreases from its maximum to about zero

at around 03:48 UTC, when the GPS station is surrounded by convective cells; it decreases to a minimum of -2.79 mm at 04:35 UTC. The modulus of the gradient vector actually shows a second maximum at the end of the motion of the convective zone through the GPS station, when, once again, the anisotropy of the troposphere around the GPS station is the most pronounced (Fig. 6d). At around 05:40 UTC, anisotropy of the stratiform part of the MCS is detected in the northeast direction by gradients, which is confirmed by reflectivity maps (Fig. 6e). Thus, the tropospheric gradients retrieved from the GIPSY-OASIS software appear to be relevant to contribute to the description of the dynamics of intense weather events such as MCSs. Tropospheric gradients retrieved once per 24 h session from the GAMIT software provide no such information, but the examination of the postfit phase residuals is more informative in this case.

### 3.3 GPS phase residuals

The daily WRMS computed by the software for the sessions covering the 11 August 2006 are 15 mm with GAMIT and 10 mm with GIPSY-OASIS (see Sect. 2.3.2). In this subsection, we analyze the behavior of the phase residuals during the case study in detail. Therefore, Fig. 9 shows the root mean square (RMS) of the GPS phase residuals computed at a 15 min sampling interval (panel a), the number of GPS observations per epoch (panel b), and the GPS phase residuals (panel c) retrieved from the GIPSY-OASIS (black) and the GAMIT (red) software packages during the MCS event.

With GIPSY-OASIS, the 15 min RMS of the GPS phase residuals exceeds 15 mm between 03:10 and 04:35 UTC during the crossover of the convective cells. This indicates that the tropospheric anisotropy is more complex than can be inferred from gradients, which is confirmed by the GPS phase residuals on the reflectivity maps (Fig. 6c). Indeed at 03:30 UTC, the GPS phase residuals exceed 25 mm in the northeast quadrant, are below -10 mm in the northwest quadrant, and are between -7 and 7 mm elsewhere. Before and after this period between 03:10 and 04:35 UTC, the RMS generally oscillates between 5 and 10 mm. Otherwise, the RMS can be slightly higher due to very few mis-modeled and non-deleted GPS observations that produce residuals which are 30 mm over the absolute value (Fig. 9c, black dots). In general, the behavior of the GPS phase residuals reflects the part of the tropospheric anisotropy that cannot be captured by the gradients. This is displayed as spatial correlations in their projected sky-plots (Figs. 6, 10) and is reflected as temporal correlations in their time series (Fig. 9c). Increased scatter in the GPS phase residuals retrieved from GIPSY-OASIS is clearly visible during the MCS event (Fig. 9c), but the 5 min sampling interval of the GPS data does not make easy to detect their temporal correlations. The number of GPS observations per epoch is usually over nine and decreases slightly during the entire event but remains above seven (Fig. 9b). The small number of observations that are rejected indicates



**Figure 9.** Outputs from GIPSY-OASIS (black) and GAMIT (red) software during the MCS event on 11 August 2016 at Niamey (Niger): (a) the 15 min RMS of GPS phase residuals (mm), (b) the number of GPS observations per epoch and (c) GPS phase residuals (mm) at full temporal resolution (5 min for GIPSY-OASIS and 30 s for GAMIT). Rainfall (mm  $h^{-1}$ ; blue bar plots) is retrieved from the ARM Mobile Facility. The times of the reflectivity maps given in Fig. 6 are shown using vertical dashed green lines and correspond to (a) 02:10 UTC, (b) 02:40 UTC, (c) 03:30 UTC, (d) 04:40 UTC, and (e) 05:40 UTC.

that the chosen settings of the GPS data processing with the GIPSY-OASIS software appear to be adequate to study this kind of extreme weather event.

With the GAMIT software, the RMS of the GPS phase residuals computed at a 15 min sampling interval oscillates between 4 and 8 mm before and after the MCS event (Fig. 9a). It skyrockets as soon as a sufficient number of GPS observations cross the MCS from 8 to 20 mm at 02:00 UTC, then it oscillates between 11 and 22 mm as long as the MCS affects the GPS data. The RMS decreases from over 20 mm at 05:25 UTC to 5 mm at 05:50 UTC while the stratiform part of the MCS is still above the GPS station. Thus, we can assume that the GPS observations that cross the convective part of the MCS are the most disrupted; hence, these observations are poorly modeled during the GPS data processing which leads to higher phase residuals and, in turn, to a higher RMS. Further understanding of the RMS variation while the MCS affects the GPS data is delicate as up to 50 % of the observations per epoch can be rejected as outliers (Fig. 9b, red line).



Figure 10. Similar to Fig. 6 but with GPS post-fit phase residuals from the GAMIT processing, with a 30 s sampling interval, over a 5 min span centered on the times of the radar plots.

This loss of data has to be put into perspective: the 30 s sampling interval of the GPS data processed by GAMIT leads to the processing of 10 times more data than with GIPSY-OASIS – thus, many more residuals are available for study (Fig. 9c, red dots). Strong deviations in the phase residuals are clearly visible during the MCS event on some satellites as revealed by temporally correlated patterns. Therefore, we examined the projected GPS phase residuals from GAMIT and superimposed them onto the reflectivity maps retrieved from the MIT C-band Doppler radar (Fig. 10).

Assuming a tropospheric thickness of 10 km and a spherical Earth with a radius of 6400 km, the GPS phase residuals are projected on a horizontal plane of the local coordinate system centered on the GPS station to be approximately superimposed on the reflectivity maps retrieved from the MIT C-band Doppler radar. The results obtained at 02:10 UTC (panel a), 02:40 UTC (panel b), 03:30 UTC (panel c), 04:40 UTC (panel d), and 05:30 UTC (panel e) are given by Fig. 6 for GIPSY-OASIS and by Fig. 10 for GAMIT.

At 02:10 UTC, the GPS phase residuals exceed 20 mm in the southeast direction because the associated GPS observations are disturbed by the top of the MCS's anvil, which can not be detected by the radar waves at an elevation of  $0.7^{\circ}$ . Between 02:10 and 03:00 UTC, the GPS phase residuals within a radius of 25 km are within an acceptable range; however, beyond 25 km, GPS phase residuals located to southeast indicate the presence of convective cells and exceed 20 mm, while those located in other directions are below -20 mm (Fig. 10b). At 03:30 UTC, convective cells are above the GPS station and GPS phase residuals are very disturbed within a radius of 25 km, while those beyond 25 km located to the east or west are below -20 mm (Fig. 10c). At 04:40 UTC, convective cells are located west of the station. The GPS phase residuals within a radius of 25 km are negative but stay in an acceptable range, whereas those corresponding to signals going through the convective cells always exceed 20 mm. At 05:40 UTC, the GPS phase residuals located to the northwest exceed 20 mm which explains the signal detected in the north component of the tropospheric gradients in the previous subsection. Thus, with the GAMIT solution, the tropospheric anisotropy induced by the MCS is mainly reflected in GPS phase residuals, and the east-west crossing of the MCS can be detected in the temporal evolution of the extreme phase residuals in the maps of the Fig. 10. However, for a more quantitative study, the tropospheric gradients retrieved by GIPSY-OASIS would be more accurate.

### 4 Composites of surface variables and GPS estimates during MCS events over West Africa

Section 3 provided a detailed analysis of a single case study. Here, we aim to characterize the West African MCS properties in a systematic way using the data from the six AMMA GPS stations over three monsoon seasons, between 2006 and 2008.

### 4.1 Detection of MCS at AMMA GPS stations

A crucial step is to define a method to properly detect MCSs based on surface meteorological data. Provod et al. (2016) used 1 min averaged pressure, temperature, and wind observations from the ARM-MF in Niamey to detect the arrival of cold pools associated with MCSs, which they subjectively verified from the MIT radar and/or Meteosat satellite images. They detected 42 cold pools during the Special Observing Period (SOP) of the AMMA project (1 June–30 September 2006). Among them, 33 were squall-line MCSs, 4 were non-squall-line MCSs, 1 was from a freshly dissipated MCS, and 4 were from local non-MCS convection.

As in the study by Provod et al. (2016), here we use a method of detecting cold pools based on surface meteorological data. These data, available from the PTU200 sensors attached to the AMMA GPS stations, include pressure, temperature, and relative humidity (recorded with a sampling rate of 15 min), but no wind data. First, we primarily detect rainfall events and then select those that show the surface air temperature drop characteristics of the arrival of cold pools ahead of the MCS's peak rain rate. The first detection step is fundamental because substantial rainfall implies the presence of significant cloud cover which attenuates the variations of the surface temperature due solar radiation and thus avoids false detections and fails due to the daily solar cycle.

The quality of the PTU200 data was first assessed by comparison with the ARM-MF data at Niamey. The mean  $\pm$  one standard deviation of differences in pressure, temperature, and relative humidity were  $0.0 \pm 0.1$  hPa,  $0.4 \pm 0.8$  °C, and  $0.6 \pm 1.9$  %, respectively. The rain rate data were retrieved from the 3-hourly,  $0.25^{\circ} \times 0.25^{\circ}$  gridded precipitation data (3B42 v6) from the Tropical Rainfall Measuring Mission (TRMM) Multisatellite Precipitation Analysis (Huffman et al., 2007). The TRMM rain rates were compared point to point to the ARM-MF data at a 3h sampling interval (2972 points). Rainfall was detected at 243 points by either TRMM or ARM-MF data but only at 74 points in both data sources. At the 169 remaining points, rainfall was observed either only by TRMM or only by ARM-MF. At 145 points from the 169 (86 %), the rain rate was less than  $0.5 \text{ mm h}^{-1}$ . The 16 remaining points (14%) corresponded to more intense rainfall, but it was observed at different times. This result led us to define a rainfall event as an uninterrupted period of precipitation during which the maximum rain rate is higher than  $0.5 \,\mathrm{mm}\,\mathrm{h}^{-1}$ .

We detect 27 events using the 3 h ARM-MF rainfall data during JJAS 2006 and 47 events using the TRMM rainfall data. Among the 27 ARM events, 25 coincided with at least one TRMM rainfall event. The two ARM events not detected by TRMM are of very low intensity and most probably correspond to small local showers. Each event is associated with a start time, a time of maximum precipitation, and an end time. If several rainfall events occur within a 13h window they are merged together as we are mainly interested in MCSs events in which several convective cells can be detected. We keep the rainfall events with average precipitation above  $0.5 \text{ mm h}^{-1}$  and with a peak rain rate above  $1 \text{ mm h}^{-1}$ . This method detects 27 events from the TRMM data and 20 events from the ARM-MF rainfall data. Compared with the first intercomparison, the agreement between the two datasets is improved when considering longer and more intense events. The seven TRMM events that do not match with any ARM-MF rainfall event may be due to the difference in the spatial representativeness of rainfall between the two datasets: very localized for ARM data and more extended for TRMM data  $(0.25^{\circ} \sim 30 \text{ km})$ . As the GPS tropospheric estimates are sensitive to weather events occurring within a radius of approximately 75 km around the station (as illustrated in Figs. 6 and 10), we actually expect better agreement with the TRMM observations than with local precipitation observations.

For the detection of MCSs, we identify the CPCT by seeking and dating the largest temperature drop over 1 h within a time window defined from the event start time minus 4 h to the event end time plus 3 h. To detect a cold pool associated with a MCS, the temperature drop must be at least 1 °C, as in Provod et al. (2016).

Figures 11 and 12 show the events detected using our method on Niamey data during August 2006 along with surface meteorological observations from the PTU200 sensor and ARM-MF data (Fig. 11) and GPS tropospheric estimates

(Fig. 12) retrieved with both software approaches (GIPSY-OASIS as black lines and GAMIT as red dashed lines). The method detects 10 MCS events based on the rain rates from TRMM data (Fig. 11c, d) and temperature from the PTU200 (Fig. 11b). In general, all of the detected MCS cases were confirmed from the reflectivity maps of the MIT C-band Doppler radar. One event (14 August 2006) is not detected in the ARM-MF rain rate data. This event is actually a propagating cold pool from a freshly dissipated MCS (Provod et al., 2016). All periods detected include a rapid increase in the wind speed associated with a sudden change of direction (Fig. 11a), which are the characteristics of the arrival of a cold pool. They also include a pronounced and sudden drop in temperature (Fig. 11b), and an increase in relative humidity (Fig. 11c) and ground pressure (Fig. 11d). This is all consistent with previous studies (e.g., Redelsperger et al., 2002). There is also a clear peak in the ZWD (Fig. 12a), a signal in the tropospheric gradients retrieved from GIPSY-OASIS (Fig. 12b, c), and an increase in the 15 min RMS of GPS phase residuals retrieved from GAMIT (Fig. 12d). Careful inspection of the wind, temperature, tropospheric gradients, and phase residuals reveals two potential events with coldpool-like signatures on 30 and 31 August that were not detected because there was no signal in rainfall. Furthermore, there is no peak in the ZWD data associated with these potential events (Fig. 12a). Inspection of the MIT radar data actually shows decaying MCSs for these two cases.

Thus, of the 42 cold pools detected and categorized by Provod et al. (2016), the detection procedure retains the 24 most intense ones including large MCSs, mainly squall-lines, over the JJAS period in 2006. The MCS detection procedure was then applied to the six AMMA GPS stations over the JJAS periods from 2006 to 2008. We note that precipitation and lightning from the ground, as well as from clouds and lightning from geostationary satellites could be added to the MCS detection arsenal. This additional information will be considered in the future.

### 4.2 Composites of meteorological variables and GPS estimates during MCS events in JJAS from 2006 to 2008

Using the detection method described in Sect. 4.1, we flagged a total of 414 MCSs passing over the six AMMA GPS stations in JJAS from 2006 to 2008. The detailed characteristics for each event (start, peak, and rainfall end times, CPCT, temperature drop, and precipitation peak and cumulative rainfall depth) are given in the Supplement. The list only includes events for which both PTU200 and GPS data were available within the 10 h windows centered on the CPCTs (i.e., a few events may not be documented because of gaps in our data).

Table 3 reports the number of events and the statistics of the surface air variables, precipitation, and ZWD recorded by the six stations. The meridian climatic gradient between the Sahara and the Guinean coast is reflected in both the number



**Figure 11.** Surface meteorological data from the ARM Mobile Facility with a 1 min sampling interval (black or green lines) at Niamey during August 2006: (a) wind speed  $(m s^{-1})$  and direction (°), (b) temperature (°C), (c) relative humidity (%), and (d) pressure (hPa). The 3 h rainfall (mm h<sup>-1</sup>; blue bar plots) is retrieved (b) from the ARM Mobile Facility and (c, d) from the TRMM Multisatellite Precipitation Analysis. Each MCSs detected is shown using two dashed vertical dashed magenta lines indicating the start and end of the rainfall period.

of MCS events and the atmospheric variables. The number of events ranges from 31 in Timbuktu to 116 in Djougou, i.e., with a ratio of 1:4. Note that the number of events detected at Tamale is smaller due to failures with the GPS receiver in 2007. The atmospheric environment of these MCSs is characterized using the surface air temperature, pressure, and relative humidity, and the ZWD. Again, the meridional climatic gradient is clearly reflected in these variables. Surface air temperature and relative humidity range from the very hot (over  $34 \,^{\circ}$ C) and dry (below  $45 \,^{\circ}$ ) Saharo–Sahelian climate to the milder (below  $30 \,^{\circ}$ C) and moister (over  $69 \,^{\circ}$ ) Sudano–Guinean climate. The mean surface pressure is not very informative here as it mainly reflects the altitude of the **Table 3.** Statistics (mean  $\pm$  one standard deviation) for 414 MCS events detected at the AMMA GPS stations during JJAS from 2006 to 2008. Characteristics of the atmospheric environment before the MCS: surface air temperature, pressure, and relative humidity from PTU200 sensors, and ZWD from GPS receivers (these variables are averaged over the 5 h before the cold pool crossing time). Cumulative rainfall depth and the maximum rain rate from the TRMM Multisatellite Precipitation Analysis (Huffman et al., 2007). The last column gives the maximum temperature drop,  $\Delta$ Temp, within the 1 h following the cold pool crossing time.

Station	Alt. (m)	Number of MCSs	Temperature (°C)	Relative humidity (%)	Pressure (hPa)	ZWD (mm)	Cumulative rainfall (mm)	Rain rate max $(mm h^{-1})$	ΔTemp (°C)
TOMB	265	31	$34.3\pm4.1$	$43.4 \pm 16.0$	$979.4 \pm 1.8$	$269.8\pm26.8$	$10.8\pm8.6$	$3.3\pm2.6$	$-8.1\pm3.4$
GAO1	260	59	$34.7\pm3.9$	$39.5 \pm 15.9$	$979.1\pm2.1$	$262.8\pm29.5$	$10.4\pm7.8$	$3.0\pm2.3$	$-8.3\pm3.8$
NIAM	224	74	$30.6\pm3.9$	$60.0\pm16.1$	$985.0 \pm 1.9$	$292.7\pm27.6$	$16.1\pm13.4$	$4.0\pm3.0$	$-6.5\pm3.2$
OUAG	307	89	$29.8\pm3.4$	$63.6 \pm 14.8$	$976.4 \pm 1.6$	$295.5\pm26.1$	$21.7 \pm 19.9$	$5.5\pm4.8$	$-6.8\pm2.9$
TAMA	172	45*	$28.6\pm3.2$	$69.3 \pm 13.0$	$991.7 \pm 1.4$	$312.4\pm20.3$	$12.6\pm7.7$	$2.9\pm1.8$	$-5.2\pm2.9$
DJOU	439	116	$26.7\pm2.4$	$74.4 \pm 11.5$	$962.6 \pm 1.3$	$292.0\pm21.3$	$15.7\pm13.0$	$3.7\pm3.3$	$-4.6\pm2.2$

\* The number of events detected at TAMA is smaller because of a long gap in the GPS data in 2007.

station, but variability is seen to be stronger at the Sahelian sites (Niamey, Gao, and Timbuktu). A contrast is also seen in the ZWD values that are below 270 mm at Timbuktu and Gao and above 290 mm at the other sites. The ZWD characterizes the total column water vapor which is, as expected, lower on average and shows a stronger variability in Mali than at the southernmost sites. However, the relative contrast between the ZWD values is not as strong as the contrast between the surface relative humidity values. Surface humidity is strongly controlled by surface evapotranspiration, which is a more limiting factor due to the relatively low soil moisture content and vegetation cover in the Sahel (Lohou et al., 2014), whereas ZWD is largely influenced by upper level moisture transport, which is more of a large-scale nature e.g., synoptic-scale fluctuations induced by African easterly waves can be quite large at the northernmost sites (Barthe et al., 2010). In the south, the lower saturation vapor pressure due to relatively cooler air is also a limiting factor to both surface humidity and the ZWD. In line with Lebel et al. (2003) and Frappart et al. (2009), rainfall characteristics also show marked latitudinal variations, with a minimum cumulative rainfall of around 10 mm at the Saharo–Sahelian stations (Timbuktu and Gao) and a maximum at the Sudano-Sahelian stations (16 and 21 mm at Niamey and Ouagadougou, respectively). Interestingly, the Saharo–Sahelian stations show the most important surface air temperature drop (around -8 °C in 1 h on average); this reflects the strong cooling (and moistening) of the cold pool air (Lothon et al., 2011; Provod et al., 2016) when it enters the typically warmer and drier boundary layer. Figures 13 and 14 show composites of the surface meteorological variables, the ZWD, and the tropospheric delay gradient components, centered on the CPCT of the MCSs for all six AMMA stations. Overall, the variables show similar temporal variations at all of the sites. The variations are also highly consistent with those observed in the 11 August case study presented in Sect. 3: a drop in surface air temperature coincident with a jump in relative humidity and in surface pressure, as well as a gradual increase in the ZWD that peaks slightly after ( $\sim 30 \text{ min}$ ) the CPCT. The east gradient component also shows a systematic oscillation across the CPCT, whereas the north gradient is more stationary throughout the time window. There is a clear increase of the magnitude of all changes around the CPCT with the latitude of the site (along the south-north climate gradient). Surface air temperature (Figs. 13a, 14a) and relative humidity (Figs. 13b, 14b) do not fluctuate much before and after the CPCT. The magnitudes of the relative humidity and surface pressure jumps are coupled with the magnitudes of the temperature drops. They reach over 30 % RH and 2 hPa at Gao and Timbuktu, compared with only 15 %-20 % and 1 hPa at Djougou and Tamale. Small trends in these variables can actually be seen after the CPCT at the four northernmost sites and reveal a gradual drying and warming of the surface air after the passage of MCSs.

Figure 14d shows that the ZWD starts to increase 5 h before the CPCT at a common rate of around  $3 \text{ mm h}^{-1}$ . In the 30 min before the CPCT, this rate increases, and it is only then that it depends on the latitude of the station. The lowest rates of  $8 \text{ mm h}^{-1}$  are found at Djougou and Tamale, and the highest rates of  $16 \text{ mm h}^{-1}$  are seen at Niamey, Gao, and Timbuktu. The ZWD peaks  $\sim 30 \text{ min}$  after the CPCT at all sites. Thus, the change in the ZWD is strongly station (latitude) dependent: it is more marked at the northernmost stations where it reaches around 40 mm (relative to a reference value taken 2 h before the CPCT); it reaches 30 mm at Ouagadougou; and it only reaches 20 mm at the southernmost sites. Hence, moisture convergence associated with the propagating MCSs is larger in the initially drier Saharo–Sahelian atmospheres than in the moister Guinean region. At CPCT plus 5 h, the ZWD has generally decreased back to or slightly below its initial value. Although there is a strong moisture convergence associated with the passage of the MCS, the tendency after it is a slightly drier air column, especially in the more arid climate (this is consistent with typical signatures found in sounding data, not shown). The ZWD peak is also relatively narrow (less than 12 h), and it is narrower from the



**Figure 12.** Tropospheric estimates from GIPSY-OASIS (black) and GAMIT (red) at Niamey during August 2006: (a) zenithal wet delay (mm), (b) east and (c) north components of tropospheric gradients (mm), and (d) the 15 min RMS of GPS phase residuals (mm). Each MCS detected is shown using two dashed vertical magenta lines indicating the start and end of the rainfall period.

southern to the northern sites, which suggests faster MCSs in the north (and is again consistent with existing studies, e.g., Maranan et al., 2018). It also implies that the moisture convergence occurs at sub-synoptic scales.

The isotropic vision of water vapor changes in the atmospheric column described by the ZWD time series is complemented by tropospheric gradients which reflect the horizontal displacement of MCSs around the GPS stations. The east component (Fig. 13e) has a very similar behavior at all AMMA GPS stations. It is very low at CPCT minus 5 h and slightly increases to show a maximum at the CPCT when the anisotropy of the troposphere around the GPS station is most pronounced. It then decreases rapidly to present a minimum during the 1–2 h following the CPCT, which corresponds to the east–west crossing of the convective part of the MCSs. After this minimum, the east component tends to return to a



Figure 13. Composites of surface air (a) temperature ( $^{\circ}$ C), (b) relative humidity (%), and (c) pressure (hPa) from the in situ PTU200 sensor, and (d) the ZWD (mm), (e) east components of the tropospheric delay gradient, and (f) north components of the tropospheric delay gradient (mm) at all six AMMA stations, in a 10h window centered on the cold pool crossing time (CPCT).

value close to zero, corresponding to the passage of the stratiform part of the MCSs. The behavior of the east component previously observed in Sect. 3.2 seems to be common to all GPS stations in West Africa during the passage of MCSs, implying that MCSs preferentially propagate in the east-west direction (which is very consistent with numerous past studies, e.g., Mathon and Laurent, 2001). Conversely, the north component of the tropospheric gradients (Fig. 13f) does not show such a systematic behavior. It would be valuable to assess whether these differences in the meridional direction agree with findings from MCS tracking (using satellite data). A stronger signal is found at Timbuktu where this component remains negative on average throughout the time window. This suggests that convective cells pass preferentially south of the station, which is a result that may be related to the location of the site, on the northern flank of the intertropical convergence zone (ITCZ). The signal is distinct and also



**Figure 14.** Same as Fig. 13 but variables are differentiated by the average values computed over the 5 h period preceding the cold pool crossing time (CPCT). The average values are given in Table 3.

weaker at all of the other stations. However, it is noticeable that it is very similar at Niamey to the one observed for the case study (Fig. 8), which suggests that convective cells have a slight tendency to arrive at the station from the southeast and then turn west.

### 5 Conclusions

The characteristics of GPS tropospheric estimates (ZWD and gradients) and post-fit phase residuals during the wet season (June to September) of the West African monsoon were evaluated using two different GPS data processing approaches. The first uses the double-difference observations from a regional network of GPS stations centered on the study area (West Africa) with the GAMIT software. The second uses the undifferenced observations in PPP mode with the GIPSY-OASIS software. Hourly ZWD estimates retrieved using both strategies are highly consistent on average (RMS difference of  $\sim 4$  mm). Both processing solutions exhibit a strong seasonal modulation of the ZWD and gradient variability reflecting the presence of active moisture transport during the monsoon period. In this respect, the 5 min sampling interval of the ZWD and gradient estimates with GIPSY-OASIS appears to be more suitable for studying rapid weather events compared with the 1-hourly ZWD and daily gradient sampling of GAMIT. Sub-daily gradient variability is shown to be quite large (1 $\sigma$  of 0.6 mm on average with peaks up to 2 mm) and to represent a strong source of uncertainty in the GAMIT estimates. Mis-modeling of the tropospheric delay variability in the GAMIT solution is reflected in the post-fit residuals during the monsoon season (the weighted RMS of residuals passes from 10 mm during the dry season to 15 mm on average during the wet season). Mesoscale convective systems (MCSs) are a major contributor to the rapid tropospheric delay anisotropy.

The case study of the squall line over Niamey on 11 August 2006 confirmed the good sensitivity of both GPS software packages to the ZWD variation associated with the passage of a MCS. In this case, a rapid increase of 40 mm was observed in the ZWD over the 2 h preceding the rainfall peak followed by a decrease leading to a slightly lower final ZWD value (20 mm below the initial value). Thanks to the high temporal sampling of the gradient estimates, the GIPSY-OASIS solution provides additional information on the atmospheric anisotropy. The east gradient component shows a strong oscillation reflecting the westward propagation of the MCS passing over the station. With the GAMIT solution, the horizontal anisotropy is reflected to some extent in the postfit residuals, as could be verified on reflectivity maps from the MIT C-band Doppler radar.

These results were extended by the analysis of composites of surface meteorological variables and GPS delay estimates at all six AMMA stations over three monsoon seasons (2006–2008). It shows a very consistent behavior in all variables during the time window of  $\pm 5$  h around the cold pool crossing times (CPCTs) of all 414 MCS events. Qualitatively, all MCS events show a drop in the surface air temperature and a jump in the relative humidity and pressure at the CPCT consistent with the squall-line case study and the results of Provod et al. (2016) for Niamey. Quantitatively, however, there is a clear tendency toward a stronger magnitude of the changes at higher latitude, i.e., a dependance on the atmospheric environment specific to each site. The dry Saharo-Sahelian climate of Gao and Timbuktu is prone to the steepest surface temperature drops  $(-9 \degree C)$  and increases in ZWD (40 mm, i.e., about  $6 \text{ kg m}^{-2}$  IWV) due to moisture convergence close to the time when the cold pool arrives. However, cumulative rainfall (10 mm) and the peak rain rate  $(3 \text{ mm h}^{-1})$  are lower in this region than at the Sudano-Sahelian sites of Ouagadougou and Niamey. This may be due to the stronger evaporation of rainfall drops (Meynadier et al., 2010a) in drier climates. Furthermore, at this scale the total column moisture shows a tendency to diminish after the passage of the MCS, indicating that the balance between moisture convergence and precipitation is negative; however, the opposite is actually seen at the moist Soudano-Guinean sites (Djougou and Tamale). Finally, the tropospheric gradients show that the main direction MCS propagation at all six sites is from east to west, which is consistent with climatology based on the satellite imagery (e.g., Laing and Fritsch, 1993).

To conclude, this study showed that the high frequency estimates of the ZWD and the tropospheric delay gradients are relevant for climate monitoring and the documentation of intense weather events such as MCSs. They could also be used to study other rapid meteorological processes and for verification and assimilation in numerical weather prediction models.

*Data availability.* GPS and PTU200 data of AMMA stations can be accessed at ftp://igs.ign.fr//pub/igs/data/campaign/amma (last access: 27 November 2018). ARM-MF data can be accessed at http://www.archive.arm.gov (last access: 27 November 2018).

*Supplement.* The supplement related to this article is available online at: https://doi.org/10.5194/acp-19-9541-2019-supplement.

*Author contributions.* SN performed the data analysis (processing of the GPS data, and the computation of statistics) and wrote the paper. OB performed the MCS event detection based on precipitation and surface meteorological data. OB and FG contributed to the data analysis and writing the paper.

*Competing interests.* The authors declare that they have no conflict of interest.

*Special issue statement.* This article is part of the special issue "Advanced Global Navigation Satellite Systems tropospheric products for monitoring severe weather events and climate (GNSS4SWEC) (AMT/ACP/ANGEO inter-journal SI)". It is not associated with a conference.

Acknowledgements. The authors would like to thank Miroslav Provod (School of Earth and Environment, University of Leeds, UK) for the details provided in his database of 42 cold pools in Niamey, Niger. ARM-MF data were obtained from the Atmospheric Radiation Measurement (ARM) program sponsored by the U.S. Department of Energy, Office of Science, Office of Biological and Environmental Research, Climate and Environmental Sciences Division. The TRMM\_3B42 data were provided by the NASA/Goddard Space Flight Center's Mesoscale Atmospheric Processes Laboratory and Precipitation Processing System, which develop and compute the TRMM\_3B42 data as a contribution to TRMM, and are archived at the NASA Goddard Earth Sciences Data and Information Services Center. This study contributes to the IdEx Université de Paris ANR-18-IDEX-0001. This work is also a contribution to the European COST Action ES1206 GNSS4SWEC (GNSS for Severe Weather and Climate monitoring; http://www.cost.eu/COST\_Actions/essem/ES1206, last access: 25 July 2019) which aims at the development of a global GPS network for atmospheric research and climate change monitoring. This is IPGP contribution number 4052.

*Financial support.* This work was supported by CNES through the TOSCA committee (grant no. 4359). It was also supported by the

Centre National de la Recherche Scientifique (grant LEFE/INSU projet VEGA).

*Review statement.* This paper was edited by Mathias Palm and reviewed by two anonymous referees.

### References

- Adams, D. K., Fernandes, R. M., Kursinski, E. R., Maia, J. M., Sapucci, L. F., Machado, L. A., Vitorello, I., Monico, J. F., Holub, K. L., Gutman, S. I., Filizola, N., and Bennett, R. A.: A dense GNSS meteorological network for observing deep convection in the Amazon, Atmos. Sci. Lett., 12, 207–212, https://doi.org/10.1002/asl.312, 2011.
- Adams, D. K., Minjarez, C., Serra, Y., Quintanar, A., Alatorre, L., Granados, A., Vázquez, E., and Braun, J.: Mexican GPS Tracks Convection From North American Monsoon, Eos Trans. AGU, 95, 61–68, https://doi.org/10.1002/2014EO070001, 2014.
- Altamimi, Z., Rebischung, P., Metivier, L., and Collilieux, X.: ITRF2014: a new release of the International Terrestrial Reference Frame modeling nonlinear station motions, J. Geophys. Res.-Sol. Ea., 121, 6109–6131, https://doi.org/10.1002/2016JB013098, 2016.
- American Meteorological Society: Mesoscale Convective System, Glossary of Meteorology, available at: http://glossary.ametsoc. org/wiki/Mesoscale\_convective\_system (last access: 27 November 2018), 2015.
- Bar-Sever, Y. E., Kroger, P. M., and Borjesson, J. A.: Estimating horizontal gradients of tropospheric path delay with a single GPS receiver, J. Geophys. Res., 103, 5019–5035, https://doi.org/10.1029/97JB03534, 1998.
- Barthe, C., Asencio, N., Lafore, J., Chong, M., Campistron, B., and Cazenave, F.: Multi-scale analysis of the 25–27 July 2006 convective period over Niamey: Comparison between Doppler radar observations and simulations, Q. J. Roy. Meteor. Soc., 136, 190– 208, https://doi.org/10.1002/qj.539, 2010.
- Bertiger, W., Desai, S. D., Haines, B., Harvey, N., Moore, A. W., Owen, S., and Weiss, J. P.: Single receiver phase ambiguity resolution with GPS data, J. Geodesy, 84, 327–337, https://doi.org/10.1007/s00190-010-0371-9, 2010.
- Beucher, F., Lafore, J., Karbou, F., and Roca, R.: Highresolution prediction of a major convective period over West Africa, Q. J. Roy. Meteor. Soc., 140, 1409–1425, https://doi.org/10.1002/qj.2225, 2014.
- Bevis, M., Businger, S., Herring, T. A., Rocken, C., Anthes, R. A., and Ware, R. H.: GPS meteorology: Remote sensing of atmospheric water vapor using the global positioning system, J. Geophys. Res., 97, 15787–15801, https://doi.org/10.1029/92JD01517, 1992.
- Bock, O., Bouin, M. N. Doerflinger, E., Collard, P., Masson, F., Meynadier, R., Nahmani, S., Koité, M., Gaptia Lawan Balawan, K., Didé, F., Ouedraogo, D., Pokperlaar, S., Ngamini, J.-B., Lafore, J. P., Janicot, S., Guichard, F., and Nuret, M.: West African Monsoon observed with ground-based GPS receivers during African Monsoon Multidisciplinary Analysis (AMMA), J. Geophys. Res., 113, D21105, https://doi.org/10.1029/2008JD010327, 2008.

- Boehm, J., Niell, A., Tregoning, P., and Schuh, H.: Global Mapping Function (GMF): A new empirical mapping function based on numerical weather model data, Geophys. Res. Lett., 33, L07304, https://doi.org/10.1029/2005GL025546, 2006a.
- Boehm, J., Werl, B., and Schuh, H.: Troposphere mapping functions for GPS and very long baseline interferometry from European Centre for Medium-Range Weather Forecasts operational analysis data, J. Geophys. Res., 111, B02406, https://doi.org/10.1029/2005JB003629, 2006b.
- Bretherton, C. S., Peters, M. E., and Back, L. E.: Relationships between Water Vapor Path and Precipitation over the Tropical Oceans, J. Climate. 17. 1517-1528, https://doi.org/10.1175/1520-0442(2004)017<1517:RBWVPA>2.0.CO:2, 2004.
- Chen, G. and Herring, T. A.: Effects of atmospheric azimuthal asymmetry on the analysis of space geodetic data, J. Geophys. Res., 102, 20489–20502, https://doi.org/10.1029/97JB01739, 1997.
- Chong, M.: The 11 August 2006 squall-line system as observed from MIT Doppler radar during the AMMA SOP, Q. J. Roy. Meteor. Soc., 136, 209–226, https://doi.org/10.1002/qj.466, 2010.
- Couvreux, F., Rio, C., Guichard, F., Lothon, M., Canut, G., Bouniol, D., and Gounou, A.: Initiation of daytime local convection in a semi-arid region analysed with high-resolution simulations and AMMA observations, Q. J. Roy. Meteor. Soc., 138, 56–71, https://doi.org/10.1002/qj.903, 2012.
- D'Amato, N. and Lebel, T.: On the characteristics of the rainfall events in the Sahel with a view to the analysis of climatic variability, J. Climatol., 18, 955–974. https://doi.org/10.1002/(SICI)1097-0088(199807)18:9<955::AID-JOC236>3.0.CO;2-6, 1998.
- Dong, D.-N. and Bock, Y.: Global Positioning System Network analysis with phase ambiguity resolution applied to crustal deformation studies in California, J. Geophys. Res., 94, 3949–3966, https://doi.org/10.1029/JB094iB04p03949, 1989.
- Dow, J. M., Neilan, R. E., and Rizos, C.: The International GNSS Service in a changing landscape of Global Navigation Satellite Systems, J. Geodesy, 83, 191–198, https://doi.org/10.1007/s00190-008-0300-3, 2009.
- Fink, A. H., Vincent, D. G., and Ermert, V.: Rainfall Types in the West African Sudanian Zone during the Summer Monsoon 2002, Mon. Weather Rev., 134, 2143–2164, https://doi.org/10.1175/MWR3182.1, 2006.
- Frappart, F., Hiernaux, P., Guichard, F., Mougin, E., Kergoat, L., Arjounin, M., Lavenu, F., Koité, M., Paturel, J.-E., and Lebel, T.: Rainfall regime over the Sahelian climate gradient in the Gourma, Mali, J. Hydrol., 375, 128–142, https://doi.org/10.1016/j.jhydrol.2009.03.007, 2009.
- Gutzler, D., Kim, H. K., Higgins, W., Juang, H., Kanamitsu, M., Mitchell, K., Mo, K., Pegion, P., Ritchie, L., Schemm, J. K., Schubert, S., Song, Y., and Yang, R.: North American monsoon Model Assessment Project: (NAMAP). NCEP/Climate Prediction Center ATLAS No. 11. National Oceanic and Atmospheric Administration, U.S. Department of Commerce, 35 pp., available at: http://www.cpc.ncep.noaa.gov/research\_ papers/ncep\_cpc\_atlas/11/index.html (last access: 27 November 2018), 2004.
- Hernandez-Pajares, M., Juan, J., Sanz, J., and Orus, R.: Secondorder ionospheric term in GPS: Implementation and im-

pact on geodetic estimates, J. Geophys. Res., 112, B08417, https://doi.org/10.1029/2006JB004707, 2007.

- Herring, T. A., King, R. W., Floyd, M. A., and McClusky, S. C.: GAMIT Reference Manual: GPS Analysis at MIT Release 10.6, report, Mass. Inst. of Technol., Cambridge, available at: http://geoweb.mit.edu/~simon/gtgk/GAMIT\_Ref.pdf (last access: 25 July 2019), 2015.
- Hinderer, J., de Linage, C., Boy, J.-P., Gegout, P., Masson, F., Rogister, Y., Amalvict, M., Pfeffer, J., Little, F., Luck, B., Bayer, R., Champollion, C., Collard, P., Le Moigne, N., Diament, M., Deroussi, S., de Viron, O., Biancale, R., Lemoine, J.-M., Bonvalot, S., Gabalda, G., Bock, O., Genthon, P., Boucher, M., Favreau, G., Séguis, L., Delclaux, F., Cappelaere, B., Oi, M., Descloitres, M., Galle, S., Laurent, J.-P., Legchenko, A., and Bouin, M.-N.: The GHYRAF (Gravity and Hydrology in Africa) experiment: Description and first results, J. Geodyn., 48, 172–181, https://doi.org/10.1016/j.jog.2009.09.014, 2009.
- Houze Jr., R. A.: Observed structure of mesoscale convective systems and implications for large-scale heating, Q. J. Roy. Meteor. Soc., 115, 425–461, https://doi.org/10.1002/qj.49711548702, 1989.
- Houze Jr., R. A.: Stratiform Precipitation in Regions of Convection: A Meteorological Paradox?, B. Am. Meteorol. Soc., 78, 2179–2196, https://doi.org/10.1175/1520-0477(1997)078<2179:SPIROC>2.0.CO;2, 1997.
- Houze Jr., R. A.: Mesoscale convective systems, Rev. Geophys., 42, RG4003, https://doi.org/10.1029/2004RG000150, 2004.
- Houze Jr., R, A., Smull, B. F., and Dodge, P.: Mesoscale Organization of Springtime Rainstorms in Oklahoma, Mon. Weather Rev., 118, 613–654, https://doi.org/10.1175/1520-0493(1990)118<0613:MOOSRI>2.0.CO;2, 1990.
- Huffman, G. J., Bolvin, D. T., Nelkin, E. J., Wolff, D. B., Adler, R. F., Gu, G., Hong, Y., Bowman, K. P., and Stocker, E. F.: The TRMM Multisatellite Precipitation Analysis (TMPA): Quasi-Global, Multiyear, Combined-Sensor Precipitation Estimates at Fine Scales, J. Hydrometeorol., 8, 38–55, https://doi.org/10.1175/JHM560.1, 2007.
- Jade, S. and Vijayan, M. S. M.: GPS-based atmospheric precipitable water vapor estimation using meteorological parameters interpolated from NCEP global reanalysis data, J. Geophys. Res., 113, D03106, https://doi.org/10.1029/2007JD008758, 2008.
- Jade, S., Vijayan, M. S. M., Gaur, V. K., Prabhu, T. P., and Sahu, S. C.: Estimates of precipitable water vapour from GPS data over the Indian subcontinent, J. Atmos. Sol.-Terr. Phy., 67, 623–635, https://doi.org/10.1016/j.jastp.2004.12.010, 2005.
- Janicot, S., Thorncroft, C. D., Ali, A., Asencio, N., Berry, G., Bock, O., Bourles, B., Caniaux, G., Chauvin, F., Deme, A., Kergoat, L., Lafore, J.-P., Lavaysse, C., Lebel, T., Marticorena, B., Mounier, F., Nedelec, P., Redelsperger, J.-L., Ravegnani, F., Reeves, C. E., Roca, R., de Rosnay, P., Schlager, H., Sultan, B., Tomasini, M., Ulanovsky, A., and ACMAD forecasters team: Large-scale overview of the summer monsoon over West Africa during the AMMA field experiment in 2006, Ann. Geophys., 26, 2569– 2595, https://doi.org/10.5194/angeo-26-2569-2008, 2008.
- Jarlemark, P. O. J., Emardson, T. R., and Johansson, J. M.: Wet delay variability calculated from radiometric measurements and its role in space geodetic parameter estimation, Radio Sci., 33, 719– 730, https://doi.org/10.1029/98RS00551, 1998.

- Kouba, J.: A guide to using International GPS Service (IGS) products. IGS Central Bureau, Pasadena, available at: https://pdfs.semanticscholar.org/eb6c/a326892e99c1567dd7e647ba78fe7735a419.pdf (last access: 25 July 2019), 2003.
- Lafore, J.-P., Flamant, C., Guichard, F., Parker, D. J., Bouniol, D., Fink, A. H., Giraud, V., Gosset, M., Hall, N., Höller, H., Jones, S. C., Protat, A., Roca, R., Roux, F., Saïd, F., and Thorncroft, C.: Progress in understanding of weather systems in West Africa, Atmos. Sci. Lett., 12, 7–12, https://doi.org/10.1002/asl.335, 2011.
- Laing, A. G. and Fritsch, J. M.: Mesoscale Convective Complexes in Africa, Mon. Weather Rev., 121, 2254–2263, https://doi.org/10.1175/1520-0493(1993)121<2254:MCCIA>2.0.CO;2, 1993.
- Laing, A. G., Carbone, R., Levizzani, V., and Tuttle, J.: The propagation and diurnal cycles of deep convection in northern tropical Africa, Q. J. Roy. Meteor. Soc., 134, 93–109, https://doi.org/10.1002/qj.194, 2008.
- Largeron, Y., Guichard, F., Bouniol, D., Couvreux, F., Kergoat L., and Marticorena, B.: Can we use surface wind fields from meteorological reanalyses for Sahelian dust emission simulations?, Geophys. Res. Lett., 42, 2490–2499,, https://doi.org/10.1002/2014GL062938, 2015.
- Larson, K. M., Small, E. E., Gutmann, E. D., Bilich, A. L., Braun, J. J., and Zavorotny, V. U.: Use of GPS receivers as a soil moisture network for water cycle studies, Geophys. Res. Lett., 35, L24405, https://doi.org/10.1029/2008GL036013, 2008.
- Lebel, T., Diedhiou, A., and Laurent, H.: Seasonal cycle and interannual variability of the Sahelian rainfall at hydrological scales, J. Geophys. Res., 108, 8389, https://doi.org/10.1029/2001JD001580, 2003.
- Lohou, F., Kergoat, L., Guichard, F., Boone, A., Cappelaere, B., Cohard, J.-M., Demarty, J., Galle, S., Grippa, M., Peugeot, C., Ramier, D., Taylor, C. M., and Timouk, F.: Surface response to rain events throughout the West African monsoon, Atmos. Chem. Phys., 14, 3883–3898, https://doi.org/10.5194/acp-14-3883-2014, 2014.
- Lothon, M., Campistron, B., Chong, M., Couvreux, F., Guichard, F., Rio, C., and Williams, E.: Life cycle of a mesoscale circular gust front observed by a C-band Doppler radar in West Africa, Mon. Weather Rev., 139, 1370–1388, https://doi.org/10.1175/2010MWR3480.1, 2011.
- Lyard, F., Lefevre, F., Letellier, T., and Francis, O.: Modeling the global ocean tides: Modern insights from FES2004, Ocean Dynam., 56, 394–415, https://doi.org/10.1007/s10236-006-0086-x, 2006.
- MacMillan, D. S.: Atmospheric gradients from very long baseline interferometry observations, Geophys. Res. Lett., 22, 1041– 1044, https://doi.org/10.1029/95GL00887, 1995.
- MacMillan, D. S. and Ma, C.: Atmospheric gradients and the VLBI terrestrial and celestial reference frames, Geophys. Res. Lett., 24, 453–456, https://doi.org/10.1029/97GL00143, 1997.
- Maranan, M., Fink, A. H., and Knippertz, P.: Rainfall types over southern West Africa: Objective identification, climatology and synoptic environment, Q. J. Roy. Meteor. Soc., 144, 1628–1648, https://doi.org/10.1002/qj.3345, 2018.
- Mathon, V. and Laurent, H.: Life cycle of Sahelian mesoscale convective cloud systems, Q. J. Roy. Meteor. Soc., 127, 377–406, https://doi.org/10.1002/qj.49712757208, 2001.

- Means, J. D.: GPS Precipitable Water as a Diagnostic of the North American Monsoon in California and Nevada, J. Climate, 26, 1432–1444, https://doi.org/10.1175/JCLI-D-12-00185.1, 2013.
- Meynadier, R., Bock, O., Guichard, F., Boone, A., Roucou, P., and Redelsperger, J.-L.: West African Monsoon water cycle: 1. A hybrid water budget data set, J. Geophys. Res., 115, D19106, https://doi.org/10.1029/2010JD013917, 2010a.
- Meynadier, R., Bock, O., Gervois, S., Guichard, F., Redelsperger, J.-L., Agustí-Panareda, A., and Beljaars, A.: West African Monsoon water cycle: 2. Assessment of numerical weather prediction water budgets, J. Geophys. Res., 115, D19107, https://doi.org/10.1029/2010JD013919, 2010b.
- Miller, M. A. and Slingo, A.: The Arm Mobile Facility and Its First International Deployment: Measuring Radiative Flux Divergence in West Africa, B. Am. Meteorol. Soc., 88, 1229–1244, https://doi.org/10.1175/BAMS-88-8-1229, 2007.
- Nahmani, S.: GPS data processing methodologies for environmental sciences: contributions to the study of West African Monsoon, PhD thesis, Univ. Pierre and Marie Curie, Paris, France, 213 pp., 2012.
- Nahmani, S., Bock, O., Bouin, M. N., Santamaría-Gómez, A., Boy, J. P., Collilieux X., Métivier, L., Panet, I., Genthon, P., de Linage, C., and Wöppelmann, G.: Hydrological deformation induced by the West African Monsoon: Comparison of GPS, GRACE and loading models, J. Geophys. Res., 117, B05409, https://doi.org/10.1029/2011JB009102, 2012.
- Petit, G. and Luzum, B.: IERS Conventions 2010 (IERS Technical Note; 36), Frankfurt am Main: Verlag des Bundesamts für Kartographie und Geodäsie, 179 pp., ISBN 3-89888-989-6, available at: https://www.iers.org/IERS/EN/Publications/TechnicalNotes/ tn36.html/ (last access: 25 July 2019), 2010.
- Petrie, E. J., King, M. A., Moore, P., and Lavallée, D. A.: Higher-order ionospheric effects on the GPS reference frame and velocities, J. Geophys. Res., 115, B03417, https://doi.org/10.1029/2009JB006677, 2010.
- Provod, M., Marsham, J. H., Parker, D. J., and Birch, C. E.: A Characterization of Cold Pools in the West African Sahel, Mon. Weather Rev., 144, 1923–1934, https://doi.org/10.1175/MWR-D-15-0023.1, 2016.
- Redelsperger, J., Diongue, A., Diedhiou, A., Ceron, J., Diop, M., Gueremy, J., and Lafore, J.: Multi-scale description of a Sahelian synoptic weather system representative of the West African monsoon, Q. J. Roy. Meteor. Soc., 128, 1229–1257, https://doi.org/10.1256/003590002320373274, 2002.
- Redelsperger, J.-L., Thorncroft, C., Diedhiou, A., Lebel, T., Parker, D. J., and Polcher, J.: African Monsoon Multidisciplinary Analysis (AMMA): An international research project and field campaign, B. Am. Meteorol. Soc., 87, 1739–1746, https://doi.org/10.1175/BAMS-87-12-1739, 2006.
- Rickenbach, T., Nieto-Ferreira R., Guy, N., and Williams, E.: Radar-observed squall line propagation and the diurnal cycle of convection in Niamey, Niger, during the 2006 African Monsoon and Multidisciplinary Analyses Intensive Observing Period, J. Geophys. Res., 114, D03107, https://doi.org/10.1029/2008JD010871, 2009.
- Risi, C., Bony, S., Vimeux, F., Chong, M., and Descroix, L.: Evolution of the stable water isotopic composition of the rain sampled along Sahelian squall lines, Q. J. Roy. Meteor. Soc., 136, 227– 242, https://doi.org/10.1002/qj.485, 2010.

#### S. Nahmani et al.: Sensitivity of GPS tropospheric estimates to mesoscale convective systems

- Rochetin, N., Couvreux, F., and Guichard, F.: Morphology of breeze circulations induced by surface flux heterogeneities and their impact on convection initiation, Q. J. Roy. Meteor. Soc., 143, 463– 478, https://doi.org/10.1002/qj.2935, 2017.
- Saastamoinen, J.: Atmospheric correction for the troposphere and stratosphere in radio ranging satellites, in: The Use of Artificial Satellites for Geodesy, Geoph Monog Series, vol. 15, edited by: Henriksen, S. W., Mancini, A., and Chovitz, B. H., AGU, Washington, D.C.. USA, 247–251, https://doi.org/10.1029/GM015p0247, 1972.
- Schiro, K. A., Neelin, J. D., Adams, D. K., and Lintner, B. R.: Deep Convection and Column Water Vapor over Tropical Land versus Tropical Ocean: A Comparison between the Amazon and the Tropical Western Pacific, J. Atmos. Sci., 73, 4043–4063, https://doi.org/10.1175/JAS-D-16-0119.1, 2016.
- Schmid, R., Steigenberger, P., Gendt, G., Ge, M., and Rothacher, M.: Generation of a consistent absolute phasecenter correction model for GPS receiver and satellite antennas, J. Geodesy, 81, 781–798, https://doi.org/10.1007/s00190-007-0148-y, 2007.
- Selle, C. and Desai, S.: Optimization of tropospheric delay estimation parameters by comparison of GPS-based precipitable water vapor estimates with microwave radiometer measurements, IGS Workshop 2016, 8–12 February 2016, Sydney, NSW, Australia, available at: http://www.igs.org/assets/pdf/ W2016-PY0504-Selle.pdf (last access: 27 November 2018), 2016.

- Serra, Y. L., Adams, D. K., Minjarez-Sosa, C., Moker, J. M., Arellano, A. F., Castro, C. L., Quintanar, A. I., Alatorre, L., Granados, A., Vazquez, G. E., Holub, K., and DeMets, C.: The North American Monsoon GPS Transect Experiment 2013, B. Am. Meteorol. Soc., 97, 2103–2115, https://doi.org/10.1175/BAMS-D-14-00250.1, 2016.
- Sherwood, S. C., Roca, R., Weckwerth, T. M., and Andronova, N. G.: Tropospheric water vapor, convection, and climate, Rev. Geophys., 48, RG2001, https://doi.org/10.1029/2009RG000301, 2010.
- Taylor, C. M., Belušić, D., Guichard, F., Parker, D. J., Vischel, T., Bock, O., Harris, P. P., Janicot, S., Klein, C., and Panthou, G.: Frequency of extreme Sahelian storms tripled since 1982 in satellite observations, Nature, 544, 475–478, https://doi.org/10.1038/nature22069, 2017.
- Zipser, E. J.: Mesoscale and Convective–Scale Downdrafts as Distinct Components of Squall-Line Structure, Mon. Weather Rev., 105, 1568–1589, https://doi.org/10.1175/1520-0493(1977)105<1568:MACDAD>2.0.CO;2, 1977.
- Zumberge, J. F., Heflin, M. B., Jefferson, D. C., Watkins, M. M., and Webb, F. H.: Precise point positioning for the efficient and robust analysis of GPS data from large networks, J. Geophys. Res., 102, 5005–5017, https://doi.org/10.1029/96JB03860, 1997.